Author’s response to reviews

Title: Severe Persistent Unremitting Dermatitis, Chronic Diarrhea and Hypoalbuminemia in a Child; Hartnup Disease in Setting of Celiac Disease

Authors:

Mutaz Sultan Dr (drmutazsultan@gmail.com)
Imad Dweikat Dr (imaddweikat@yahoo.ca)
Mohammad Awar Dr (moh_4u4ever@yahoo.com)
Maher Shahrour Dr (mr_shahrour@hotmail.com)
Bassam Abu Libdeh Dr (drbassam@hotmail.com)
Thomas Ciecierega Dr (ciecierega@gmail.com)

Version: 4
Date: 29 August 2014

Author’s response to reviews:

Good day:

All the requested formatting changes have been addressed including the manuscript body, competing interest, author’s contribution and acknowledge.